Author's response to reviews

Title: Building bridges between doctors and patients: the design and pilot evaluation of a training session in argumentation for chronic pain experts

Authors:

Claudia Zanini (claudia.zanini@paraplegie.ch)
Piercarlo Sarzi-Puttini (piercarlo.sarziputtini@gmail.com)
Fabiola Atzeni (atzenifabiola@hotmail.com)
Manuela Di Franco (Manuela.Difranco@uniroma1.it)
Sara Rubinelli (sara.rubinelli@paraplegie.ch)

Version: 6
Date: 1 April 2015

Author's response to reviews:

Dear Editors,

Thank you for accepting our manuscript after the substantial revisions we made. As requested, I reordered the numbering of tables and made the necessary changes in the text, so that the tables are now chronologically mentioned in the text.

Best regards

Claudia Zanini